Dec 10th, 2007

Kent, Jon

Replaced 7500 on tower.  Swapped cards on towertop logger and installed new card.  Replaced N2 gas on closed path system.  

Dec 5th, 2007

Kent, Jon Boro

Swapped cards on all loggers, found the towertop had stopped logging for about a day.  
November 9, 2007

Chris, Jon

Vh = 0 to 0.5 ms-1

Slight drizzle

MF towertop: took Li-7500 sensor head down for maintenance; disconnected power and SDM com cables in Li-7500 box, and commented Li7500 sampling command in program. Only after disconnecting the SDM cable, the CSAT was reading reasonable values.

MF subcan: uploaded new loggerprogram that now powercycles NL100 at midnight and noon to reboot it.

Swapped cards on MF towertop, and MF subcan

Tested scheduled FTP with bat files: worked just fine. Transfer successful.

November 8, 2007

Calibrated all of the 7500’s

Towertop
Zero h20 reading -2.23, Set to zero

H20 dew point reading 5.01, Set to 5.00

Zero co2 reading 20.02, Set to zero

Span co2 reading  408.38, Set to 402.33

7000 at towertop (Closed Path)
Did calibrations under positive pressure with pump turned off

H20 zero good

Span h20 reading 4.8, set to 5.03

Co2 zero reading 0.1, used make cell B match cell A command

Span C02 reading 401.7, set to read 402.33

With pump running at 4 lpm

Span reading 403.3 ppm 

Zero co2 reading 0.5ppm

Subcan

Zero h20 reading 0.23, set to 

H20 span reading 4.92, set to 5.02

Zero co2 reading, -1.70, set to zero

Span reading 400.00, set 402.33

November, 6, 2007

Kent, Jon

Confirmed that the reference gas was not a problem by piping the caltank N2 through both cells then using the reference tank and the caltank.  The results were the same.  

Calibrated 7000 for N2 and span C02.  

After we did the N2 the span seemed to be very close.  Water channel seemed to be right on for zero water, so did not do anything for it.  

Replaced filter at inlet of tube.
Replaced reference cell gas (2000 psi)

Swapped cards on all loggers 

Removed the midcan EC system.
Noted tank pressures on co2 profile system.  

November, 5, 2007

Kent, Jon, Andrea

Replaced 7000 in box, but appears to still be about 11 ppm high for N2 and 20 ppm high on span.  
October, 31, 2007

Kent, Jon 

Removed Li_7000 from tower top.  Replaced desiccant in detector head and ran the 7000 in the lab for a few days.  Replaced battery in HOBO 2m system and downloaded data.  It should be logging now. 
October 25. 2007

Chris, Ray

Downloaded HOBO-Temp profile ( only 2m probe failed, rest was successful

Cleaned radiation domes, were okay

Increased N2 flow through Li7000 up to 0.25 lpm
October 17, 2007

Kent, Jon

Swapped cards on all loggers.  Installed long term tank on profile system (401.92). Installed new N2 for closed path system. It was found to not be empty.  
October 3, 2007

Kent, Jon 
Calibrated the closed path.  Swapped card on towertop logger.  

September 20, 2007

Chris

Installed new N2 tank for Li7000 (previous was completely empty)

Reset modem via Ace 3G ( works fine now

Swapped cards on all systems ( okay

Li7000: CO2 concentrations were fairly low (around 344 ppm) ( recalibrate!
September 12, 2007

Chris

Hooked up soil Cr10x to remote link

N2 tank for Li7000 has only 500 psi left
August 27, 2007

Kent, Jon Boro

Swapped cards on all loggers at site.

Calibrated subcanopy 7500

N2 zero reading 0.63

Set to zero

Span reading about 10.00 set to 10.01

Co2 zero reading  0.77

Set to zero

Span co2 reading 423.83 set to 429.81

Calibrated Mid tower system 

N2 zero reading 1.13

Set to zero

H20 reading 11.05

Set to 10.02

CO2 reading 428.8

Set to 429.81

Also calibrated closed path system but did not get numbers for this instrument.

Changed N2 for closed path system and changed two of the gases on profile system and found that the valve on the lowest level had been turned off.  
August 10, 2007

Chris, James

Remote link: Continued installing equipment, an additional NL100 now sits in subcanopy flux box. Hooked up Cr5000 Handararray to network, disconnected Power 10X (not needed at the moment). Soil logger not connected yet, need longer & different cable. Set com speed of all ports (CS I.O and Rs232) to 9600. Configured router for remote management (u: admin, p: 15416026889-mf).

Started time sync of all CR5000s (towertop,midcan,subcan,handarray) via remote link at 16:45 PST
August 3, 2007

Kent, Chris

Swapped all cards in loggers

Installed remote link equipment in shed (15416026889.internet.mycingular.com).
For profile system pressures are:

367=450psi

324=1650

400=200psi

457=600psi

N2=1400psi

Subcan is 6 secs slow= set to kents computer

Handar is 5 secs slow = set ot kents computer

Towertop is 5 secs slow= set to kents computer

Midcan is 5 secs slow = set to kents computer

Profile is 19 secs slow= set to kents computer

From shed to subcan is 30m

And from shed to james logger is 54m

And from shed to handar logger is 20m

July, 23, 2007

Chris

T ~ 20 deg C, vh : 2-3 ms-1, 7/8

Checked towertop system. Swapped cards. Data look good, flow through Li-7000 is about 7 lpm. 

MetOne wind sentry doesn’t seem to work (no wind speed data, but wind direction).

Downloaded complete temp profile on tower (HOBOs).

July, 19, 2007

Kent, Jon Boro

Put closed path back up, sample is set to 90.0kpa with 7.5l/m flow and n2 is set to 0.1l/m

Changed filter on the inlet and the internal filter for closed path  
A few wiring changes where made-print new map on server
Swapped cards on all loggers and set time to laptop

10mflux was 15 sec slow

Top was 16 secs slow

Handar was 15 secs slow

Subcan was 18 secs slow

Could not get cup wind speed to work on logger.

Downloaded the top hobo only-forgot to do the rest.

Looked over the profile system but could not see any problems.
June 21, 2007

Kent, Jon boro

Nice windy day.

Swapped cards on all loggers except the Handar logger.  Synced all loggers except profile system Cr1000 which was only a few secs fast.

Towetop was 21 secs fast

Midcan was 8 secs slow

Subcan was 11 secs slow

Handar logger was 8 secs slow 

Calibrated 7500 at tower top

Zero h20 reads .50 set to zero

Span h20 reads 6.12 set to 6.18

Zero co2 reads -1.43 set to zero

Span co2 reads 431.20 set to 429.81

Calibrated subcan system

Zero h20 reading 0.23 set to zero

Span reading 8.37 set 6.75

Zero co2 reading -1.64 set to zero 

Span co2 reading 431.25 set to 429.81

Found the pump in the closed path to be burned out.  Could not get it going. Will have to  replace it soon.  

Commented out the LI_7000 variables out of program.

Could not calibrate midcan because of lack of tubing to reach this instrument.  Will have to install Tees and valves to get this done. 
June 5, 2007

Kent 
Swapped cards on Handar logger and updated time on all loggers

Towertop was right on

Midcan was 2 secs fast

Subcan was right on

Handar was 8 secs fast

May 30, 2007

Kent,

Swapped cards on all loggers except handar logger

Swapped N2 tank on closed path system.  There was about 250psi left.
May 24, 2007

Kent, Chris

Warm, very nice spring day

Replaced LI_7500 head 0066 with 0557. uploaded calibration coefficients (sensor was lab calibrated in morning with fresh scrubber bottles). Replaced cable from electronics box to sensor head.   Seemed to fix problem.

Installed last Handar.

Installed cameral tripod on tower and tested fog machine.

N2 for closed path system at about 800 psi.  
May 21, 2007

Chris
Vh ~ 3.5 ms-1, T ~ 10.5 C, 7/8, rainy

Handar-array

Installed 3 additional handars recorded through SDI between 60 and 90 m apart from existing network. One more SDI-handar needs to be mounted and started. Updated program on CR5000, now all sensors (handars analog + SDI) are queried and logged every 3 sec, plus data are block vector-averaged over 12 sec. Started collecting data 16:58 PST. Logger clock NOT synchronized.

Towertop

Checked Li-7500: data look bad (AGC 94%, co2 and h2o conc unreasonable). Needs to be looked at, probably exchanged with okay probe. Other sensors look good. N2 pressure is 1000 psi. Logger clock NOT synchronized.

Midcan

System looks good. Logger clock NOT synchronized.
May 9, 2007

Kent, Chris,

Nice day-warm

Swapped N2 tanks for closed path system

New tank at 2250psi

Swapped cards and updated time on the following loggers:

Profile 2 secs slow

Midcan  2 secs slow

Towertop  3 secs slow

Handar logger  2 secs slow 

Subcan 2 secs slow

All seem okay (CT: are the tower loggers okay?)

Something wrong with the 7500 on the towertop-check data 

CT unplugged and plugged the 7500, it seems to be working when we left.  

May 4, 2007

Kent,

Much nicer day- no rain, but cool.

Installed a serial cable from mid-canopy system to the shack.

Started mid-canopy system

Set all CR5000 clocks to my laptop

Mid-canopy system was right on

Towertop system was 15 sec fast

Handar system was 13 sec fast

Subcan system was 12 sec fast

N2 was at 600 psi

The midcan csat boom is at 151 degrees
Need to swap cards on the flux systems next week and N2 tank.
May 3, 2007

Kent, Chris,

Windy, rainy, all out a nasty spring day.

Installed the Mid-canopy system (CSAT, LI_7500, NR lite, K&P Par sensor), but did not get it running because of a bad card.

Need to read field notebook for instrument heights

From notes:  separation distance from 7500 to csat: 7 in

EC system is at 16.4 M

NRlite and Par at 15.9 M

N2 tank for closed path system was at 700 psi 
April 23, 2007
Kent

Fixed bug in towertop CR5000 program (previous used since Mar 16, 07 did not record Li-7000 water vapour time series, DOY 075 - 113)
April 19, 2007

Kent, Chris, Jon

Weak winds, temps ~ 12 deg C, no rain, cloudy

Calibrated Li7000 and Li7500 at towertop with dewpt gen + cal gases: both sensors were off by up to 10 ppm in both zero and span, details see site log book
Swapped cards
Synchronized logger clocks (towertop,handararray,subcanopy): were ALL 9 sec slow! Must have been offset in time on Kent’s laptop. All data for last download should be fine and synchronized

Swapped Gelman filter in CP box: flow jumped up from 7 to 8 lpm

Installed new N2 tank, adjusted the N2 ref gas flow to 0.125 lpm

April, xx

April, xx

KENT: here is your input required, you installed CR5000 for handararray, and went back one more time to synchronize loggers
March 30th
Chris, James, Jon

Winds 4 – 6 ms-1, Air temp ~ 12 deg C, cloudy

Handararray: collected data, card reader again ceased operation after ~ 4 hrs. Cr23x data logger clock was 2 sec off compared to laptop, did not synchronize (battery died); checked orientation of 3 vertical Handars: were approx 20 deg E of N ( corrected 

Towertop:

Swapped cards, okay, data seem to look okay.

Installed inlet filter (Gelman 1.0 um) on Li-7000 inlet, did not replace filter in enclosure (need to do this, is fairly black) ( pressure dropped by only 0.4 kPa to 90.3 kPa

Subcanopy:

Swapped cards, okay, data seem okay.

Next time:

· bring new Gelman 1.0 PTFE with barbed hose connectors

· Bolt for vertical MetOne boom

· Handararray: replace Cr23x with Cr5000 or switch to 10 sec run avg samples in 23x memory

March 28th
Chris

Weak winds (0-2 ms^-1), temps in subcanopy ~ 8 deg C, sunny

Rearranged Handar-array: Handar 2,3,6 are now installed on an additional 10’ subcanopy handar tower. Didn’t touch Handar 1,4,5. New installation heights are:

Handar2 : 1.0 m agl

Handar3: 1.9 m agl

Handar6: 3.45 m agl

Additional subcanopy handar tower is located on a straight line between old Handar2 and Handar3 locations (hor. Distance 7.9 m), 3.15m away from the old Handar3 location towards old Handar3. 

Checked Cr23x logger clock: showed 03/28/07 01:04:50 AM, laptop was 03/28/07 12:00:36 PM PST ( ~ 11 hrs slow! When did this happen???? ( synchronized logger time with laptop; started collecting data ~ 12:05 PST

No compass ( Handars 2,3,6 and tower approx aligned with N-S, check next time out there

Checked tower logger clock: showed 12:15:40 PM, laptop was 12:15:13 PM PST ( 27 sec fast! ( logger clock synchronized w/o swapping cards! Will result in doubled records! Data of towertop logger look okay.

Checked subcanopy clock: showed 12:22:00, laptop was 12:20:53 ( 67 sec fast! ( synchronized logger lock w/o swapping cards! Data look good.

Runtime generator 188.7 hrs

Propane tank is down to 25%

N2 for Li-7000 still has 500 psi left

March 16th

kent, chris

put met one (wind sentry) back up on a center pole

59” above 38m mark ( new installation height of MetOne wind sentry  39.5 m
Changed program so that it averages the Li temp of cell every 10 secs and uses that to control the heater and fan.  

Also put in average for atm sensor 

Also reinstalled program on Handar profile logger- it seemed to be locked up. It is currently storing to the cardreader.

March 15th
Kent, Chris,

March 15th
Clear, sunny

Subcan-

Calibrated 7500

Zero H20- reading 0.32-set to zero

Span h20 reading 10.97- set to 13.23

Zero co2 reading -2.98 set to zero zero

Span co2 reading 427.99 set to 429.81

Removed Q7.1

Installed BF3 diffuse radiation
Installed 7500 back on track with fresh desiccant

Removed met one to fit for boom

February 19, 2007

Kent, Chris
Installed new logger program on Cr5000 towertop: now includes correct thresholds for CNR-1 KUp and KDown variables

Permanently installed serial cable for data communication between towerbase and towertop logger, cut a hole in loggerbox and sealed with putty. Took down short-haul modems.
Swapped cards on Cr5000 towertop 
February 15, 2007

Kent, Chris
Swapped N2 tanks for Li-7000, pressure is 2100 psi. New tank will last for ~ 20 days!

Uploaded new logger program to Cr5000 towertop: variables in datatable tempconf get read only every 10 sec (used tobe 1 sec) to relax stress on fan/ heat strips.

Swapped cards on Cr5000 towertop 
Took down Li-7500 head (around 13.40 PST) for desiccant replacement, logger box still hooked up, Li-7500 data recorded as 0s. 
February 08, 2007

Chris, Jon

Quick check only: Li-7000 seems to be fine. N2 pressure is 1050 psi ( consumption is about 100 psi/ day

Swapped cards on Cr5000 towertop, logger clock okay!
February 02, 2007

Kent, Chris
Towertop: Calibrated Li-7500 and Li-7000 for CO2, NO water vapour

Swapped cards, installed new logger program on CR5000 towertop
January 30, 2007

Kent, Chris
Reinstalling CP-system: runs just fine. Li-7000 has not been calibrated in-situ!

Cleaned Li-7500 windows, diagnostic value is now back to 249 (was 250).
Installed new logger program on Cr5000 towertop
Swapped cards on subcanopy system (card was filled), misunderstanding lead to complete loss of subcan data since last download!

New CSAT direction from Jan17, 2007 on: 350 deg!
Installed K&Z CNR-1 4-component net radiometer
January 19, 2007

Kent, Chris, Jon

Reinstalling electronics box in CP-system: no success, take down entire CP system.
January 17, 2007

Kent, Chris

Partly cloudy, no wind, thawing

Installed new 6 AWG copper power cable on tower for CP system ( works fine, took down aluminum power cable.

Emergency power shut-off occurred at around 1:40 PM. Started generator manually, had to dry off the screen of pressure valve. Had to do it repeatedly when starting the generator.

Power supply for MKS pressure controller of CP system failed ( need to troubleshoot, CP system still not running as no flow through system

Installed short-haul modems for CR5000 towertop ( works fine at 9600 Baud rate.

Replaced CSAT on towertop with rain-upgraded one (ser# 355). Changed SDM address of CSAT in logger program to 2. Need to check orientation of new sonic next time out there (should be approximately the same as before).
Installed new towertop CR5000 logger program: MF_CLSD_MOD.cr5.

Disconnected LI-190 PAR sensor.

January 12, 2007

Kent, Chris

Clear skies, temps around -3 C, no wind

Installed DCtoDc converter in CP system ( didn’t help, CP still powercycles when pump etc are turned on. CP still not running

Display of Cr5000 at towertop completely black ( logger is still running. Tried RS232 communication over 40m cable ( no problem!

Voltage in powergrid low, generator needs to come on soon

Installed new N2 tank for CP ( initial pressure 2100 psi

January 5, 2007

Kent, about 2:30

Cold and a little rain

Swapped cards on the Handar profile but think I got the whole 1G file instead of just the data-will have to find some way of deleting the NANs from the file.

Swapped cards on the profile system and installed a new half-hour tank.  Also raised pressure that the system operates at from what appeared to be about 102 to about 106kPa.
